Author’s response to reviews

Title: Retinal blood flow is increased in type 1 diabetes mellitus patients with advanced stages of retinopathy

Authors:
Hoang-Ton Nguyen (ht.nguyen@vumc.nl)
Eelco van Duinkerken (e.vanduinkerken@vumc.nl)
Frank. Verbraak (f.d.verbraak@amc.uva.nl)
Bettine Polak (bcp.polak@ziggo.nl)
Peter Ringens (p.ringens@mumc.nl)
Michaela Diamant (m.diamant@vumc.nl)
Annette Moll (a.moll@vumc.nl)

Version: 2  Date: 10 May 2016

Author’s response to reviews:

BMC Endocrine Disorders
Editorial board
Amsterdam, 10 May, 2016

Dear dr. Clunn,

We are very pleased to hear our manuscript BEND-D-15-00073 entitled ‘Retinal blood flow is increased in type 1 diabetes mellitus patients with advanced stages of retinopathy’ has been potentially accepted for publication after several requests have been addressed.

In our revised version, we have addressed all requests:
1. An availability of data and materials section has been added.

2. The abstract section has been formatted as requested.

3. Figure legends have been moved after the References section.

4. One version of the manuscript has been uploaded instead of multiple versions.

No other changes to the manuscript have been made. We hope that all the adjustments are acceptable to the Editor.

Sincerely yours,

On behalf of all authors,

Hoang-Ton Nguyen, MD

VU University Medical Center

Department of Ophthalmology

Van der Boechorststraat 7 – Room G417

1081 BT Amsterdam

The Netherlands

Tel: +31-20-4448420

Fax: +31-20-4448230

E-mail: ht.nguyen@vumc.nl